Author's response to reviews

Title: Process of distant lymph node metastasis in colorectal carcinoma: Implication of extracapsular invasion of lymph node metastasis

Authors:

Takaaki Fujii (ftakaaki@med.gunma-u.ac.jp)
Yuichi Tabe (1@ac.jp)
Reina Yajima (2@ac.jp)
Satoru Yamaguchi (3@ac.jp)
Soichi Tsutsumi (4@ac.jp)
Takayuki Asao (5@ac.jp)
Hiroyuki Kuwano (6@ac.jp)

Version: 4 Date: 24 May 2011

Author's response to reviews:

Editorial office
BMC Cancer

Article title: Process of distant lymph node metastasis in colorectal carcinoma: Implication of extracapsular invasion of lymph node metastasis
MS ID: 6257686604812903
Authors: Takaaki Fujii, et al.
Journal: BMC Cancer

Very many thanks for considering our above manuscript.

We here send you the revised version of the manuscript, entitled “Process of distant lymph node metastasis in colorectal carcinoma: Implication of extracapsular invasion of lymph node metastasis” by Takaaki Fujii, et al.

According to your suggestions, the manuscript has been revised.
The authors would provide our best regards to your excellent reviewers for their kind and constructive comments.

This revised manuscript was read and approved the submission by all coauthors. This work has not been published and is not being considered for publication elsewhere, in whole or in part in any language.

With best regards,
Takaaki Fujii, M.D., Ph.D.